Author's response to reviews

Title: Design of a multicentre randomized trial to evaluate CT colonography versus colonoscopy or barium enema for diagnosis of colonic cancer in older symptomatic patients: The SIGGAR study.

Authors:

Steve Halligan (s.halligan@ucl.ac.uk)
Richard J Lilford (r.j.lilford@bham.ac.uk)
Jane Wardle (j.wardle@ucl.ac.uk)
Dion Morton (dion.morton@uhb.nhs.uk)
Pauline Rogers (pauline.rogers@uclh.nhs.uk)
Katherine Wooldrage (katherine.wooldrage@cancer.org.uk)
Rob Edwards (rob.edwards@cancer.org.uk)
Reshma Kanani (reshma.kanani@cancer.org.uk)
Urvi Shah (urvi.shah@cancer.org.uk)
Wendy Atkin (wendy.atkin@cancer.org.uk)

Version: 2 Date: 11 October 2007

Author's response to reviews:

Thank you for your email dated September 28th telling me that the manuscript was acceptable for publication in principle.

I am very sorry for the delay but I have at last found some time necessary to add a statement on Competing Interests and Author Contributions (both now listed after the Acknowledgements), and upload it all.

Regards,
Steve Halligan